Author’s response to reviews

Title: High prevalence of multidrug-resistant tuberculosis in Johannesburg, South Africa: a cross sectional study

Authors:

Lee Fairlie (nee Kleynhans) (leef@witsecho.org.za)
Natalie C Beylis (Natalie.Beylis@nhls.ac.za)
Gary Reubenson (Gary.Reubenson@wits.ac.za)
David P Moore (David.Moore@wits.ac.za)
Shabir A Madhi (madhis@rmru.co.za)

Version: 5 Date: 17 January 2011

Author’s response to reviews: see over
17 January 2011

Dear Editor

Dr Robert Wallis

BMC Infectious Diseases

Thank you for accepting our article for publication.

Please note the following changes to manuscript MS 1655627184428250. High prevalence of multi-drug resistant tuberculosis in African children in a setting of high HIV prevalence.

Title change: High prevalence of multidrug- resistant tuberculosis in Johannesburg, South Africa: a cross sectional study. (as per reviewer 1)

Further changes as per reviewer2:

Methods: 3\textsuperscript{rd} paragraph, 5\textsuperscript{th} sentence added.

Results: Drug susceptibility testing: 3\textsuperscript{rd} sentence added.

Discussion: 5\textsuperscript{th} paragraph: 3\textsuperscript{rd} and 4\textsuperscript{th} last sentences added.

\textbf{Major revisions:}

Author list has been changed.

Affiliation addresses have been added in full.

Abstract to be updated in the submission system. Appears within article text.

Reference 1: functioning link provided.

Figure 1: Title added: “Children with culture positive TB- drug susceptibility patterns”

2\textsuperscript{nd} paragraph of discussion, 1\textsuperscript{st} sentence, reference 14 added.

References updated to correct format.

Tables have been added to text and removed as figures.

\textbf{Minor revisions:}

Tracked changes, comments and highlighted text removed.

Title page removed.
Corresponding author added.

Methodology changed to methods.

Author contributions changed to paragraph format.

Conclusion heading added.

Please contact me should any further changes be required.

Kind Regards

Dr Lee Fairlie

0027827809997

leef@witsecho.org.za